Author’s response to reviews

Title: Simultaneous resolution of focal segmental glomerulosclerosis associated with chronic lymphocytic leukaemia treated with fludarabine, cyclophosphamide and rituximab

Authors:

Spyridon Arampatzis (spiros.arampatzis@gmail.com)
Nikolaos Giannakoulas (ngiannakoulas@hotmail.com)
Vassilios Liakopoulos (liakopul@otenet.gr)
Theodoros Eleftheriadis (teleftheriadis@yahoo.com)
Panagiota Kourtis (giotakourtis@yahoo.gr)
Foteini Karasavvidou (fotkarasa@yahoo.gr)
Panagiota Matsouka (pmatsouka@med.uth.gr)
Ioannis Stefanidis (stefanid@med.uth.gr)

Version: 2 Date: 31 March 2011

Author’s response to reviews: see over
Dear Editor,
Please find enclosed a case report manuscript entitled “Simultaneous resolution of focal segmental glomerulosclerosis associated with chronic lymphocytic leukemia treated with fludarabine, cyclophosphamide and rituximab” by S. Arampatzis et al..
We present a rare case of CLL-associated focal segmental glomerulosclerosis with nephrotic-range proteinuria. Fludarabine, cyclophosphamide and rituximab treatment was successful in inducing effective remission of CLL and resolution of nephritic-range proteinuria. Our case throws additional light on the treatment of CLL-associated nephrotic syndrome.
We would greatly appreciate it if you would consider this manuscript for publication.
None of the materials have been published previously and the manuscript is not under consideration elsewhere.
All authors read and approved the final version.
There was no funding and there are no conflicts of interest.

Yours sincerely

for the authors
S. Arampatzis